Title: Surveillance recommendations based on an exploratory analysis of respiratory syncytial virus reports derived from the European Influenza Surveillance System

Authors:

 Tamara J Meerhoff (t.meerhoff@nivel.nl)
 Douglas Fleming (dfleming@rcgpbramresunit.nhs.uk)
 Ann Smith (ann.smith@hps.scot.nhs.uk)
 Anne Mosnier (amos@openrome.org)
 Arianne B van Gageldonk-Lafeber (Rianne.van.gageldonk@rivm.nl)
 John W Paget (j.paget@nivel.nl)
 On behalf of the EISS RSV Task Group (t.meerhoff@nivel.nl)

Version: 4 Date: 4 August 2006

Author's response to reviews:

RE: MS: 1984189643106009 Surveillance recommendations based on an exploratory analysis of respiratory syncytial virus reports derived from the European Influenza Surveillance System

Dear Editor,

Thank you for accepting our manuscript for publication. We have addressed the formatting changes that were requested.

With best wishes,
Tamara Meerhoff